Author's response to reviews

Title: The positive association between number of children and obesity in Iranian women and men: Results from the National Health Survey

Authors:

Enayatollah Bakhshi (Bakhshi@razi.tums.ac.ir)
Mohammad Reza Eshraghian (eshraghian@yahoo.com)
Kazem Mohammad (mohamadk@tums.ac.ir)
Abbas Rahimi Foroushani (rahimifo@tums.ac.ir)
Hojat Zeraati (zeraatih@tums.ac.ir)
Akbar Fotouhi (afotouhi@tums.ac.ir)
Fraidon Siassi (siassif@sph.tums.ac.ir)
Behjat Seifi (behins@yahoo.com)

Version: 8 Date: 17 May 2008

Author's response to reviews:

Dear,
The BioMed Central Editorial Production Team,
Thank you for consideration of our manuscript for publication in your valuable journal.
We have reviewed the above manuscript according to your comments.

best wishes
M. R. Eshraghian
Correspondence Author

Author's response to the BioMed Central Editorial Production Team
Manuscript ID 7214024591761460
Title: The positive association between number of children and obesity in Iranian women and men: Results from the National Health Survey
Authors:
Enayatollah Bakhshi(Bakhshi@razi.tums.ac.ir)
Mohammad Reza Eshraghian(eshraghian@yahoo.com)
Kazem Mohammad(mohamadk@tums.ac.ir)
Abbas Rahimi Foroushani(rahimifo@tums.ac.ir)
Hojat Zeraati(zeraatih@tums.ac.ir)
Akbar Fotouhi(afotouhi@tums.ac.ir)
Fraidon Siassi(siassif@sph.tums.ac.ir)
Behjat Seifi(behins@yahoo.com)

Version: 6 Date: 17 May 2008

Author's response to the BioMed Central Editorial Production Team

Formatting changes requested

-----------------------------

1. Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

* Change made as indicated by the Editorial Production Team.

2. ABSTRACT - Please change this to abstract.

* Change made as indicated by the Editorial Production Team.

3. Abbreviations - Please include these as paragraph, not a list, and change the name of this section to 'Abbreviations'.

* Change made as indicated by the Editorial Production Team.